Author's response to reviews

Title: Typhoid fever as a cause of opportunistic infection: case report

Authors:

Claudia Colomba (claudia.colomba@libero.it)
Laura Saporito (laura.sapo@tin.it)
Laura Infurnari (infurnari.laura@libero.it)
Salvatore Tumminia (claudia.colomba@libero.it)
Lucina Titone (titonel@unipa.it)

Version: 2 Date: 15 August 2005

Author's response to reviews:

We made some changes in the "Conclusions" section and added some references to expand the review of previous published cases of typhoid fever.